Author's response to reviews

Title: Measuring changes in perception using the Student Perceptions of Physician-Pharmacist Interprofessional Clinical Education (SPICE) instrument

Authors:

Joseph A Zorek (jzorek@pharmacy.wisc.edu)
Eric J MacLaughlin (eric.mclaughlin@ttuhsc.edu)
David S Fike (dfike@uiwtx.edu)
Anitra A MacLaughlin (anitra.mclaughlin@gmail.com)
Mohammed Samiuddin (mohammed.samiuddin@ttuhsc.edu)
Rodney B Young (rodney.young@ttuhsc.edu)

Version: 2 Date: 29 April 2014

Author's response to reviews:

I uploaded a cover letter dated 4-29-14 that contains point-by-point responses to reviewers’ comments.